Author’s response to reviews

Title: Measuring abortion-related mortality: challenges and opportunities

Authors:

Caitlin E Gerdts (gerdtsc@obgyn.ucsf.edu)
Ozge Tuncalp (tuncalpo@who.int)
Heidi B Johnston (johnstonh@who.int)
Bela Ganatra (ganatrab@who.int)

Version: 4 Date: 31 July 2015

Author’s response to reviews:

The references have been revised and updated as requested:
Reference 3 = completed the reference
Reference 5 = last page was added
Reference 16 = It is only one page, so removed the hyphen
Reference 6 = looked up the study and added the journal information
Reference 7 = added the initial, additional authors and issue number
Reference 8 = added initial and page number
Reference 14 = added the link but do not know the access date
Reference 17 = added the link but do not know the access date
Reference 19 = changed the reference to the Lancet version of the article.